Dear editor,

Please find attached the revised figures of the manuscript “Evaluation of two formulations of adjuvanted RTS,S malaria vaccine in children aged 3 to 5 years living in a malaria-endemic region of Mozambique: a Phase I/II. Randomised double-blind bridging trial ”, in response to the reviewer 2 comment:

**Comment,**

“Minor Essential Revisions (such as missing labels on figures, or the wrong use of a term, which the author can be trusted to correct)
The formatting of the text in Figures 1 and 2 was off and should be corrected”

**Answer,**

The figure format was corrected on the current version to PNG and presented on PowerPoint.

I hope these changes are appropriate and the paper can be accepted.

Yours sincerely,

Eusebio Macete

Eusebio Macete, MD, MPH
eusebiomacete@yahoo.com

Centre de Salut Internacional
Hospital Clinic/IDIBAPS
Calle Rosselló 132, 4º
08036 Barcelona, Spain